Author's response to reviews

Title: Haplotype frequencies in a sub-region of chromosome 19q13.3, related to risk and prognosis of cancer, differ dramatically between ethnic groups

Authors:

Mikkel H Schierup (mheide@daimi.au.dk)
Thomas Mailund (mailund@birc.au.dk)
Heng Li (liheng@genomics.org.cn)
Jun Wang (wangj@genomics.org.cn)
Anne Tjønneland (annet@cancer.dk)
Ulla Vogel (ulbvo@food.dtu.dk)
Lars Bolund (bolund@humgen.au.dk)
Bjørn A Nexø (nexo@humgen.au.dk)

Version: 4 Date: 16 February 2009

Author's response to reviews:

To the Editor

We have addressed the required changes to the manuscript as follows:

1) A methods section has been included in the abstract.
2) The reference to trial registration has been removed from the abstract, and a paragraph concerning ethical approval has been included in the Methods section.
3) The Methods section has been moved before the results section.
4) The numbering of the references has been changed accordingly.

Yours sincerely

Bjorn Nexø